Weekly Status Week of May 1 – May 7, 2008
Day of Year 122 - 128
Mission Day 2726 - 2732
Earth Observing-One (EO-1) General

There were 114 Data Collection Events (DCEs) scheduled this week.  
On May 6, a telecon was held with the International Federation Red Cross (IFRC) and other international organizations working in the area of global flood disaster alerts.  The following significant items were discussed:

· IFRC was asked to send a set of coordinates for a desired target in a devastated area in Myanmar for which EO-1 was tasked to take an image on May 7.
· Ben Burford, from JAXA, is to be asked to send the IFRC any available ALOS products of pertinent areas in Myanmar before and after the recent catastrophe produced by cyclone Nargis.  
· IFRC is interested in obtaining satellite images of Myanmar that show the flood receding.
· Bob Adler’s group has enhanced their TRMM global flood map product with regional displays (e.g., for Myanmar sector http://trmm.gsfc.nasa.gov/publications_dir/hydro_model_2.html) that show an 

animated movie of a 7-day flood inundation.

· Michael Burnett was requested to provide any pertinent images contained in the ECHO (Earth Observing System Clearinghouse) database.
· The IFRC originally contacted local Red Cross representatives in and around Myanmar on April 23 concerning potential flooding based on Columbia University data that exhibited an anomalous rainfall event.  A timeline is being constructed for the period of April 23 – May 9 that will show the contacts made by the IFRC in Geneva and the data product(s) they used to formulate each notification contact.
INSTRUMENTS
All instruments operated nominally this week.
EO-1 Spacecraft Subsystems
Command and Data Handling (C&DH)

Continued to experience problems playing back engineering data from its Solid-State-Recorders.
Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

Sensor Web & Virtual Observatory Demonstrations
The weekly teleconference between the Advanced Information System Technology (AIST) Sensor Web collaborators was not held on Thursday, May 1 due to scheduling conflicts.

Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week.
EO-1 MISSION OPERATIONS CENTER
Real-Time
Spacecraft state of health is currently nominal.
Chassis current anomaly:
· A meeting was held on Friday May 2, 2008 to discuss possible troubleshooting and analysis options.

· FOT currently is scheduling TDRSS contacts during the expected times of the chassis spikes to capture more real-time data.

· FOT so far has conducted 6 TDRSS contacts as the spacecraft enters daylight.

· Chassis current remains fairly steady between -0.003 and 0.003 amps.

· FOT has not been able to capture another incident.

· FOT plans to use a modified version of the solar array characterization test to study the behavior of the incidents.

· FOT currently suspects the electric wiring of one of the segments incurs a short when the solar array is rotated to a certain position (approximately 86 deg. of a 120 deg. rotation pattern). 

· FOT has built a new Relative Time Sequence (RTS) definition, to investigate this anomaly that will be up-linked to the spacecraft.

· During the commanding sequence, solar array segments will be power cycled to possibly identify if a particular segment wiring bundle is in fact incurring a short and causing the chassis current anomaly.   

Mission Planning
No change from last report.

Flight Dynamics
No change from last report.

Data Analysis and Trending

No change from last report.

System Administration
Systems upgrades and IT security requirements:

· FOT management has provided, to the EO-1 project, an estimate of labor and cost associated with making a transition from unsupported operating systems and operational tools currently used by the FOT.

· EO-1 ground systems technology engineer will initiate upgrades to the front end data systems (FEDS) and the Level 0 processing system.

· Systems will be upgraded from UNIX to LINUX environments.

· Project will be presented with hardware and software specs required for this transition.

Operational Enhancements Requests and Discrepancy Reports
No change from last report.
GROUND AND SPACE NETWORK

Station Downtime
No major problems to report.
Operational Discrepancies
No major problems to report.
Support Enhancements  
Certification of PF-2:
· Date for the connectivity and data flow test has been moved to a TBD date past May 19, 2008.
Upcoming Events
A pair of MLT control inclination maneuvers is scheduled to be executed.

· First maneuver will be performed on May 27, 2008.

· Second maneuver will be performed on May 29, 2008.
Imagery Status
Scenes and Engineering Cals planned for week of May 1 – May 7, 2008              114
Total scenes and engineering calibrations planned for entire mission – approximately 39,663
Total Scenes:  ALI scenes in the Level 0 archive            36,047 (as of May 7, 2008)

                        Hyperion scenes in the Level 0 archive    35,795
Publications and Presentations Status (as of 12-05-07)
404 publications 

284 external presentations

53 articles and press releases
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